Author’s response to reviews

Title: Cost-effectiveness of six strategies for Helicobacter pylori diagnosis and management in uninvestigated dyspepsia assuming a high resource intensity practice pattern

Authors:

Kyland P Holmes (kyland.holmes@gmail.com)
John C Fang (John.Fang@hsc.utah.edu)
Brian R Jackson (brian.jackson@aruplab.com)

Version: 5 Date: 29 November 2010

Author’s response to reviews:

I changed the following typographical items:

* Italicized H. pylori in the abstract
* Changed "Secondorder" to "Second-order" on page 4. If the editors prefer, the hyphen could be replaced by a space.
* Changed "longterm" to "long-term" on page 5. Again, if the editors prefer, this hyphen could be replaced by a space.
* Changed "costeffectiveness" to "cost-effectiveness" on page 5; could be a space if preferred.
* Removed unnecessary line breaks from table legend on page 10.